Author's response to reviews

Title: Duodenal carcinoma at the ligament of Treitz. A molecular and clinical perspective.

Authors:

Peter T Kalogerinis (pkalogerinis@student.methodist.edu)
John E Poulos (poulosj@pol.net)
Andrew Morfesis (Morfesis@owendrivesurgical.com)
Anthony Daniels (Tdani52291@aol.com)
Stavroula Georgakila (stageorga@yahoo.com)
Thomas Daignault (poulosj@pol.net)
Alexandros G Georgakilas (georgakilasa@ecu.edu)

Version: 5 Date: 16 September 2010

Author's response to reviews:

Dear Editors,

In this 3rd revision, we have corrected the additional points you raised:

1. The authors' list and initials in the submission site.
2. New Table exists at the end of the manuscript. We hope we corrected this error since it is not clear to use what seems to be the exact problem. Table though was redone.

Thank you for acceptance and your help in the editing and correction of our manuscript.

Alexandros Georgakilas, PhD